Author's response to reviews

Title: Infections following leeches' application: a case series and review of the literature.

Authors:

Benjamin Maetz Dr. (ben.maetz@wanadoo.fr)
Ralph Abbou Dr. (Ralph.Abbou@chru-strasbourg.fr)
Jean Baptiste Andreoletti Dr. (ANDREOLETTI.J.ch-mulhouse@apicrypt.fr)
Catherine Bruant Rodier Pr. (Catherine.Bruant-Rodier@chru-strasbourg.fr)

Version: 5 Date: 20 July 2012

Author's response to reviews: see over
Dear reviewer,

According to your kind advices, we have changed several details in our manuscript.
We changed the title.
We removed the dates and numbers less than ten were converted into words.
We presented the two clinical cases in the abstract as they were presented in the article.

Sincerely yours,
Benjamin Maetz, MD